\begin{table}%T5
  \caption{\label{tab:fiducial}Properties of the fiducial model combining the parameter values and
  assumptions going into the NFW modelling.}
  \small%\centering
\par
 \begin{tabular}{ccc}
 \hline\hline\noalign{\smallskip}
   Parameter & Value & See Sect. \\ 
   \hline
   $\max(|\varepsilon|)$ & $0.8$ & \ref{sec:veri} \\
   $\min(\nu)$ & $4.5$ & \ref{sec:veri} \\
   $\min(\tr\tens{P}^{{\rm g}})$ & $0.1$ & \ref{sec:veri} \\
   $m_{{\rm bright}}$ & $20.0$ & \ref{sec:colcolsel} \\
   $m_{{\rm faint}}$ & $22.5$ & \ref{sec:colcolsel} \\
   centre & from $S$-statistics & \ref{sec:cent} \\
   radial fit range & $0\arcmin<\theta<15\arcmin$ & \ref{sec:fits} \\ 
   \noalign{\smallskip}
   $f_{0}$ & $1.08\pm0.05$ & \ref{sec:veri} \\
   $f_{1}(\theta)$ & $f_{{\rm cg}}^{{\rm NFW}}(\theta)+1$ & \ref{sec:fcg} \\
   $\langle D_{{\rm ds}}/D_{{\rm s}}\rangle$ & $0.33\pm0.03$ & \ref{sec:ddsds} \\
    \hline
 \end{tabular}
 \end{table}